Author's response to reviews

Title: Cardiac arrest associated with Sildenafil ingestion in a patient with an abnormal origin of the left coronary artery: Case report

Authors:

Bruno C Huber (bruno.huber@med.uni-muenchen.de)
Franz von Ziegler (franz.ziegler@med.uni-muenchen.de)
Fabian Bamberg (fabian.bamberg@med.uni-muenchen.de)
Wolfgang-Michael Franz (wolfgang.franz@med.uni-muenchen.de)
Alexander Becker (alexander.becker@med.uni-muenchen.de)

Version: 3 Date: 1 March 2011

Author's response to reviews: see over
Re: Manuscript # 1514453859512893

Dear Dr. Shipley,

Thank you very much for your help to improve the manuscript. As suggested we changed the statements regarding sudden cardiac death in the abstract and the manuscript to “…cardiac arrest associated with Sildenafil ingestion…”.

As before we found your comments very helpful and have revised the manuscript accordingly.

Best regards

A. Becker MD

Department of Cardiology
Ludwig-Maximilians-University, Munich
D-81377 Munich
Germany
Tel: +49 89 7095-2229
Fax: +49 89 7095-2227
Email: alexander.becker@med.uni-muenchen.de